Author’s response to reviews

Title: A cross-sectional study on peripheral arterial disease in Sri Lanka: prevalence and associated factors

Authors:

Janaka Weragoda (janaka.weragoda@gmail.com)
Rohini Seneviratne (srdeas@gmail.com)
Manuj Weerasingha (manujchri@gmail.com)
Mandika Wijeyaratne (mandika59@hotmail.com)
Anil Samaranayaka (anilsamaranayake@yahoo.com)

Version: 4 Date: 15 May 2015

Author’s response to reviews:

To improve the style of written English of the manuscript the service of Edanz Group Global Ltd was obtained. The following was made:

The Title of the manuscript was change as “A cross-sectional study on peripheral arterial disease in Sri Lanka: prevalence and associated factors